Mauna Loa Sol ar Observatory Cbserver’'s Log

Wed Sep 21 16:58:59 GMI 1994
Year: 94 Doy: 264
onserver: yasukawa

VEATHER COMMENT: Wed Sep 21 16:59: 09 GMI 1994
Cool, in noderate cirrus overcast, |ight southeast wind. Tenp=46F

DPMON COMMENT: Wed Sep 21 17:00: 15 GMI 1994
Done and Pl essey power |eft on overnight. Coude bench was left in
slew node so it did not rotate, fortunately.

Wed Sep 21 17:16:11 GMI 1994: Patrol Start

WEATHER COMMENT: Wed Sep 21 17:16:19 GMI 1994
In hole on cirrus, nay be clear

WEATHER COMMVENT: Wed Sep 21 17:28: 06 GVI 1994
Thin cirrus in the hole. WIIl not start MKIII at this tinme.

DPMON COMMENT: Wed Sep 21 17:46:53 GMI 1994
Forgot to turn clock drive for Coude bench back on after trinmng angle.

WEATHER COMMENT: Wed Sep 21 17:51:31 GMI 1994
Crrus thick again. Stopping DPMON

Wed Sep 21 17:52:02 GMI 1994: Patrol End

COVMMENT: Wed Sep 21 18:17:16 GMI 1994

Cl osing done shutter so that | can do some paint scraping while
cirrus overcast continues. WII keep one eye on the sky for
possi bl e holes in the cirrus.

COMMVENT: Wed Sep 21 22:34: 04 GMI 1994
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Chu relieving Yasukawa as observer
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WEATHER COMMENT: Wed Sep 21 23:24:19 GMI 1994
Cirrus appears to be thinning. Reopening dome. WII begin taking



data to tape. Holes in cirrus may pass into FOV in a while.

Wed Sep 21 23:24:44 GMI 1994: Patrol Start
Thu Sep 22 00:01: 03 GMI 1994: Calibration

WEATHER COMMVENT: Thu Sep 22 00: 04:16 GMI 1994

O ographi ¢ cl ouds now noving upslope in Wst. WII| pass into
FOV a fewmnutes. Still waiting for a hole in cirrus |arge
enough to get MKIIIl coronal scan

WEATHER COMMENT: Thu Sep 22 00:10: 32 GMI' 1994
O ogr aphi ¢ cl ouds now bl ocki ng FOV.

COWMENT: Thu Sep 22 00:25:17 GMI 1994

St oppi ng obs at this tinme. In heavy orographic and cirrus cloud
cover. Appears it will remain so for quite a while. WII take
this opportunity to begin painting Alice' s fanbus porch

Thu Sep 22 00: 27: 04 GMTI 1994: Patrol End

WEATHER COMMENT: Thu Sep 22 02:27:10 GMI 1994

O ographi ¢ clouds have receded. Cirrus still lingering. 1In the
thinnest portion of cirrus now WIIl restart Dpnon to see how thi ngs
| ook.

Thu Sep 22 02:27:44 GVIT 1994: Patrol Start
Thu Sep 22 02:30: 03 GMI 1994: Patrol End

COWENT: Thu Sep 22 02:33:04 GVII 1994
Qops! Dpnon doned out. WII Reconfigure dome slot. And restart
agai n.

Thu Sep 22 02:36:24 GMI 1994: Patrol Start

MKI'II COWMMENT: Thu Sep 22 02:45:51 GVII 1994
Loaf of bread diagramfinally returning to normal

WEATHER COMMENT: Thu Sep 22 02:50:29 GMI 1994
In a hole in cirrus.

MKI 'l COWMENT: Thu Sep 22 02:56: 01 GMVI' 1994
Crrus degrading | oaf of bread again. Managed to get 2 or 3
coronal scans with no cirrus interference! Yipee!



Thu Sep 22 03:00:58 GV 1994: Calibration

COMMENT: Thu Sep 22 03:30: 55 GMI 1994

Managed to obtain only a few good MKIIl and Dpron scans for
the day. Cirrus has been present to varying degrees all day,
except for several mnutes around 2:45 UT. PAU.

Activity Report:
@QP: 50-62; 89; 116; 290-304.
No Coronal Activity.

Tapes:
MIT1: HO0845
DPMON:  P00165
LOWL: LO0107 running in unit O

Thu Sep 22 03:36: 03 GMI 1994: Patrol End
Thu Sep 22 03:37:48 GMI 1994: Fil emark
SCAN- LOG

SCAN- LOG 17:15:49. 9/21/94 DOY 264
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